Author's response to reviews

Title: Morphologic and immunophenotypic evidence of in-situ Kaposi sarcoma

Authors:

Panagiotis A Konstantinopoulos (pkonstan@bidmc.harvard.edu)
Bruce J Dezube (bdezube@bidmc.harvard.edu)
Liron Pantanowitz (Liron.Pantanowitz@bhs.org)

Version: 5 Date: 28 October 2006

Author's response to reviews:

Dear Editor,

We have made the formatting revisions to our manuscript as requested. Figure 2A and 2B were changed to 2 and 3 respectively and all necessary changes to reflect this were made.

Thank you very much,

The authors